

Author's response to reviews

Title: Establishing confidence in the output of qualitative research synthesis: the ConQual approach

Authors:

Zachary Munn (Zachary.Munn@adelaide.edu.au)

Kylie Porritt (kylie.porritt@adelaide.edu.au)

Craig Lockwood (craig.lockwood@adelaide.edu.au)

Edoardo Aromataris (edoardo.aromataris@adelaide.edu.au)

Alan Pearson (alan.pearson@adelaide.edu.au)

Version: 6 **Date:** 15 August 2014

Author's response to reviews:

The editors requested a further copyedit of this paper prior to publication. We have employed the services of a professional copyeditor to review this manuscript and have also reviewed the paper once again as an author group. We have made a number of grammatical changes to this paper following this review and believe the written English is now at a level suitable for publication in a high quality peer-reviewed journal.